ORIGINAL REPORT

FACTORS ASSOCIATED WITH WORK ABILITY FOLLOWING EXERCISE INTERVENTIONS FOR PEOPLE WITH CHRONIC WHIPLASH-ASSOCIATED DISORDERS: SECONDARY ANALYSIS OF A RANDOMIZED CONTROLLED TRIAL

Hiu Kwan LO, BPhty¹, Venerina JOHNSTON, PhD, BPhty^{1,2}, Maria LANDÉN LUDVIGSSON, PhD, PT³⁻⁵, Gunnel PETERSON, PhD, PT³, Thomas OVERMEER, PhD, PT^{6,7}, Michael DAVID, PhD, MBiostat, MEd, MEpi, MSc, BSc(Hons), BEd, DipEd^{8,9} and Anneli PEOLSSON, PhD, PT³

From the ¹Division of Physiotherapy, School of Health and Rehabilitation Sciences, The University of Queensland, ²Recover Injury Research Centre, The University of Queensland, Brisbane, Australia, ³Department of Medical and Health Sciences, Division of Physiotherapy, Linköping University, Linköping, ⁴Rehab Väst, County Council of Östergötland, Motala, ⁵Department of Rehabilitation and Department of Medical and Health Sciences, Linköping University, ⁶Physiotherapy Department, School of Health, Care and Social Welfare, Mälardalen University, ⁷Centre for Health and Medical Psychology, Örebro University, Sweden and ⁸School of Public Health, The University of Queensland, Brisbane, ⁸School of Medicine and Public Health, The University of Newcastle, Newcastle, Australia

Objective: To investigate the efficacy of exercise interventions and factors associated with changes in work ability for people with chronic whiplash-associated disorders.

Design: Secondary analysis of a single-blind, randomized multi-centre controlled trial.

Setting: Interventions were conducted in Swedish primary care settings.

Patients: A total of 165 individuals with chronic whiplash-associated disorders grade II-III.

Methods: Participants were randomly allocated to neck-specific exercise, neck-specific exercise with a behavioural approach, or prescribed physical activity interventions. Work ability was evaluated with the Work Ability Index at baseline, 3, 6 and 12 months. Results: The neck-specific exercise with a behavioural approach intervention significantly improved work ability compared with the prescribed physical activity intervention (3 months, p = 0.03; 6 months, p = 0.01; 12 months, p = 0.01), and neck-specific exercise at 12 months (p=0.01). Neck-specific exercise was better than the prescribed physical activity intervention at 6 months (p = 0.05). An increase in work ability from baseline to one year for the neckspecific exercise with a behavioural approach group (p < 0.01) was the only significant within-group difference. Higher self-rated physical demands at work, greater disability, greater depression and poorer financial situation were associated with poorer work ability (p < 0.01).

Conclusion: This study found that neck-specific exercise with a behavioural approach intervention was better at improving self-reported work ability than neck-specific exercise or prescribed physical activity. Improvement in work ability is associated with a variety of factors.

Key words: employment; whiplash injuries; exercise; persistent neck pain.

Accepted May 31, 2018; Epub ahead of print Aug 8, 2018

J Rehabil Med 2018; 50: 828-836

Correspondence address: Venerina Johnston, Division of Physiotherapy, School of Health and Rehabilitation Sciences, The University of Queensland, Brisbane, Australia. E-mail: v.johnston@uq.edu.au

LAY ABSTRACT

Most people who have sustained a whiplash injury return to, or remain in, work. However, some people with whiplash injury experience ongoing problems, with difficulty performing pre-injury work duties. This study investigated the efficacy of 3 different exercise interventions on self-reported work ability for individuals with chronic whiplash. Compared with prescribed physical activity, a 3-month neck-specific exercise intervention with a behavioural approach (including education on managing relapses and stress) was significantly better at improving work ability immediately after the intervention, and at 6- and 12-month follow up. The neck-specific exercise intervention with a behavioural approach was also more effective than neck-specific exercise alone after 12 months. Poorer work ability was associated with high physical demands at work, disability, depression and poorer financial situation. Physical therapists can improve work ability for individuals with chronic whiplash through a specific exercise regime with a behavioural approach.

Whiplash-associated disorders (WAD) following road traffic accidents are a common cause of persistent pain and disability (1), posing a significant burden on society (2, 3). There is also a financial burden on the individual with WAD in terms of long-term sick leave (4), slower and lower return-to-work rates than individuals sick-listed due to other musculoskeletal disorders (5) and lower employment opportunities compared with the general population (6).

Published rates of return to work (RTW) after WAD vary: from 44% (7) to 70% (8) after one year. Discrepancies in rates of RTW are probably due to heterogeneous study populations, varying legislation and benefits, definitions of RTW and study design. Nevertheless, the majority of individuals with a whiplash injury do RTW, although approximately half will continue to experience symptoms (9). These persisting symptoms are likely to impact on work, requiring a period of modified duties and/or reduced hours. Buitenhuis et al. (10) found that, of those in paid work after a WAD (total n=733), 58% continued to be work disabled at one year. Similarly, Elfving et al. (11) showed that, while individuals with chronic WAD increased their working time after a 4-week rehabilitation programme, 62% continued to report problems with heavy work and lifting. A systematic review (12) of the effects of individual and work-related factors on work ability, as defined by the Work Ability Index (WAI), concluded that poor work ability was associated with a lack of leisure-time vigorous physical activity, poor musculoskeletal capacity, older age, obesity, high mental work demands, lack of autonomy, poor physical work environment, and high physical work load. Thus, interventions that address these factors are needed for individuals with WAD.

To date, most interventions considering work outcomes in people with WAD have been observational cohort studies (11, 13–15). The few randomized trials conducted in which work outcomes were reported included a heterogeneous sample of individuals with acute (16-18) or subacute WAD (19). No randomized trials were found that included individuals with chronic WAD. The interventions tested showed small or no improvement in work outcomes. Outcomes that have been used previously, such as RTW (yes/no) or duration of sick leave are not appropriate, as they are tied to the current social security system, employer or country. A measure such as the WAI better reflects an individual's ability to meet the occupational demands of their job and enables comparisons between studies, organizations and countries.

The current study is a secondary analysis of data to compare the efficacy of a neck-specific exercise intervention with (NSEB) and without (NSE) a behavioural approach, and prescribed physical activity (PPA), on self-reported work ability for individuals with chronic WAD grade II–III over a 12-month period. We hypothesized that participation in the NSEB would demonstrate a greater improvement in work ability compared with the other interventions. A secondary aim is to determine which of several individual, psychosocial, work and condition-specific factors is associated with changes in work ability following exercise interventions in chronic WAD.

METHODS

Study design

This prospective randomized controlled trial was registered at ClinicalTrials.gov (NCT01528579) (20). The sample size was calculated based on the primary outcome measure (7% change in Neck Disability Index) (21). Assessors were blinded. However, participants and treating physical therapists could not be blinded due to the nature of the treatment. Informed consent to participate in the study was obtained before randomization.

Setting and participants

The intervention was delivered by experienced physical therapists in the Swedish primary-care setting. Potential participants who had sought medical assistance for WAD from Swedish healthcare in 6 counties were recruited from February 2011 to May 2012, until the calculated sample size was achieved.

Participants aged between 18 and 63 years, diagnosed with WAD grade II/III, with persistent pain and disability associated with WAD that occurred 6–36 months before recruitment, who reported a score of at least 20% on the Neck Disability Index and/or 20 mm on a pain visual analogue scale (VAS) were eligible. Participants were excluded if they had persistent symptoms due to other neck trauma, previous neck surgery, spinal infection, tumour, myelopathy, signs of traumatic brain injury, a more dominant or generalized pain in other body regions, serious neck pain causing absence of work for more than one month in the previous year, or other conditions that would limit full participation in the study, such as drug abuse, a severe psychiatric disorder or limited ability to communicate in Swedish.

Potential participants were invited by post (n=7,950). Those who responded and met the self-reported inclusion criteria received a phone interview to confirm their eligibility (n=419), followed by a physical examination conducted by a physical therapist to confirm WAD grade (n=216). Specific for this secondary analysis, only those who were currently employed, or who were employed at the time of their accident, constituted the sample population (n=65) (Fig. 1). Further details of participant recruitment are provided in Ludvigsson et al. (20).

Randomization

Participants were randomly assigned to receive NSE, NSEB or PPA. An independent researcher allocated participants into different groups using a computer-generated list compiled by a statistician. Individual results were placed in opaque envelopes and sent to the independent treating physical therapists.

Interventions

The interventions (published in detail elsewhere) (20) were led by physical therapists who received one day of standardized practical and theoretical training from project leaders prior to commencement of the study. Guidance from project leaders was also provided as required. Participants were encouraged not to receive other physical treatment for their neck problem during the 3-month intervention. On completion, participants were encouraged to continue with the prescribed exercises.

Neck-specific exercise. The NSE intervention was supervised by physical therapists and performed twice weekly for 3 months. Patients were encouraged to practice the exercises at home. Education on the musculoskeletal system of the neck and information about WAD were provided. Patients were encouraged to avoid pain aggravation during the exercises. The exercises were low load initially then gradually progressed, with increased repetitions and resistance (22). Postural control was also introduced to facilitate deep cervical muscle function. On completion, a written exercise programme and prescribed general exercises were provided to encourage participants to continue those exercises.

Neck-specific exercise group with behavioural approach. The exercise component of the NSEB intervention was the same as in the NSE group, with the addition of a behavioural component. The concept of graded activity was used to educate patients not



Fig. 1. Flow diagram of participants through the study. NSE: neck-specific exercise; NSEB: neck-specific exercise with a behavioural approach; PPA: prescribed physical activity intervention; VAS: visual analogue scale; NDI: Neck Disability Index; WAD: whiplash-associated disorder.

to focus on temporary increases in neck pain. As WAD has a strong psychosocial component and several psychological and social work factors contribute to work ability (23), a behavioural approach was added to moderate the detrimental effect of some psychological factors. The behavioural component included education on the basic neuroscience of pain and how to manage pain, including relapses. Patients were also taught relaxation exercises and to decrease stress and muscle tension. Moreover, participants were empowered to set specific, measurable, achievable, realistic and time-bound (SMART) goals and progress exercise with guidance from the physical therapist. Similar to the NSE group, participants were encouraged to continue exercise after the intervention period.

Prescription of physical activity. In the PPA group, a motivational interview and physical examination was conducted in the first week by a physical therapist to inform an individualized physical activity plan. It aimed to increase the level of physical activity to improve fitness, with participants expected to complete the prescribed exercise outside the Swedish healthcare system or at home. No neck-specific exercises were prescribed. Participants

were encouraged to receive either a phone call or attend a follow-up visit to monitor their progress, but this was optional (20).

Outcome measure

The WAI was used to evaluate self-reported work ability and has been demonstrated to be a reliable and valid tool to predict work disability and retirement (24). It consists of a 7-item self-assessment of: current work ability compared with lifetime best, work ability in relation to the physical and mental demands of the job, number of current diagnosed diseases, estimated physical work impairment due to disease, sick leave during the last year, self-rated prognosis of work ability 2 years from now, and mental resources. The total score, between 7 and 49, was used to categorize participants into poor (7-27), moderate (28-36), good (37-43) and excellent (44-49) work ability (25). WAI scores were recorded at baseline before group allocation, and at 3, 6 and 12 months after commencement of the intervention. The change in WAI scores was included as the dependent variable in all analyses.

Independent variables

Potential explanatory variables assessed at baseline include: demographic (age in years, education level, financial situation, smoking status, body mass index), individual, psychosocial, work- and condition-specific, exercise factors and beliefs. Current employment status and employment status prior to their accident was determined by a series of questions about their main source of income prior to and after their WAD (paid employment, retirement, sick leave, parental leave, student aid, unemployment benefits, social security benefits, other). If not working because of their injury, the date of the last working day was recorded. Further details of the included measures are provided in the protocol for this trial (21).

Individual factors: EuroQol Five Dimension Scale and EuroQol VAS were used to measure health-related quality of life. Self-reported financial situation (good/normal/bad) and whether patients were worried about their financial situation were collected, as socio-economic inequalities have been shown to be significantly associated with work ability (26).

Psychosocial factors. Pain catastrophizing was evaluated with the Pain Catastrophizing Scale. The Tampa Scale of Kinesiophobia was used to measure fear of movement. Anxiety and depression was evaluated with the Hospital Anxiety and Depression Scales. Self-efficacy of performing daily activities with pain was rated with the 20-item Self-Efficacy Scale (0–200).

Work-specific factors. Flexibility of the work place was rated trichotomously (never/sometimes/always) for the possibility of: obtaining help from colleagues, choosing work tasks, having longer breaks at work than usual, working at home and working undisturbed. Change of work tasks, satisfaction with work tasks and the workplace were recorded dichotomously (yes/no). Perceived neck load at work was also measured with

Journal of Rehabilitation Medicine

Journal of Rehabilitation Medicine

JRM

one question: "Do you consider the load on your neck at work is: light, moderate or heavy". The Borg scale (6–20) was used to indicate the intensity of physical demands at work. A binary differences (time

to indicate the intensity of physical demands at work. A binary scale (hardly ever/frequently) was used to measure the frequency of having arms above shoulder height for more than 30 min. The Effort-Reward Imbalance Scale was used to measure work-related stress. The type of employment was also recorded (full-time/part-time).

Condition-related factors. WAD grade (II/III) and time since the accident (months) were recorded. Treatment received after their WAD injury, consultation with a physical therapist, other health professionals and use of pain medication (yes/no) were recorded. The presence of previous neck pain, back pain and current arm symptoms were also recorded dichotomously (yes/ no). The frequency of symptoms was measured dichotomously (occasionally, frequently) for neck pain, arm pain, headache, paraesthesia, neck stiffness, nausea and dizziness. Frequency of problems with swallowing, jaw, hearing, vision, sleeping, concentrating and arm lifting were also measured dichotomously. A VAS was used to record neck pain and pain bothersomeness (0-100 mm, 100=worst pain, and extremely bothersome). The level of disability due to chronic pain and neck pain were measured with the Pain Disability Index (score range 0-70) and Neck Disability Index (NDI, 0-100%), respectively.

Exercise factors and beliefs. Exercise Self-Efficacy Scale was used to rate participants' confidence on doing exercise (0–100) (27). Activity level was evaluated with International Physical Activity Questionnaire (low/moderate/high) (28). The expected treatment result was recorded on a 4-point scale (fully recovered/ substantial improvement/some improvement/no expectation).

Intervention adherence

Adherence during the active intervention period was recorded by the physical therapists in the NSE and NSEB groups and from exercise diaries of participants in the PPA group after the 3-month intervention. Maintenance of the exercise intervention after completion of the intervention at the 6- and 12-month follow-ups was determined by self-report response to one question. Exercise adherence during the active participation and maintenance phase was recorded as greater or less than 50%.

Data analysis

The analysis was performed with an intention-to-treat approach. Descriptive statistics used to summarize all included baseline variables were one-way analysis of variance for continuous variables and χ^2 tests for categorical variables. Generalized estimating equations (GEEs) were used to investigate whether a statistically significant effect of the interventions on WAI was evident over time. GEEs were used to account for correlation between participants after repeated measures, all variations in missingness of data, and unevenly spaced observations in longitudinal data. A Gaussian distribution, identity-link function GEE model was applied in the analysis and adjusted for significant factors of association with the changes in WAI scores (dependent variable). All independent variables were tested for their association with the dependent variable through a series of univariate analyses. Variables were retained if significant at 5% and model correlation structure was assessed using the Quasilikelihood Information Criterion (QIC), a version of the Akaike Information Criterion (AIC). Interactions between independent variables (group × time, group × age and group × sex) were taken into consideration. Sex, age, group, time and group × time interaction were forced into the final model. Pairwise comparisons were used to compare between (intervention groups) and within differences (time points). STATA version 14 (Stata Corporation, College Station, TX, USA) was used for all data analysis and p < 0.05 was accepted as statistically significant, with no adjustments for multiple comparisons.

RESULTS

Participants

Of the 216 participants randomized, 165 met the employment inclusion criteria and completed the WAI and were thus included in the study (Table I). Overall, baseline background characteristics between the 3 intervention groups were similar. There were no significant differences in mean WAI score at baseline (35.3, 36.0 and 34.9, p=0.64) and percentage of female participants (73.3%, 66.7% and 52.1%, p=0.07) in the 3 intervention groups, NSE, NSEB and PPA, respectively. However, participants in the NSE and NSEB groups were significantly younger than those in the PPA group (mean age 39, 40 and 44 years, respectively, p=0.03). More participants in the NSE intervention had received treatment after WAD than in the NSEB and PPA groups (p=0.01). The mean and SD for the factors significantly associated with WAI in the final model are shown in Table I. The mean WAI raw values and the relative percentage change from baseline at each time-point for each group are shown in Table II. There was a relative increase in work ability from baseline for participants in both exercise groups, with a corresponding decrease in the PPA group. Participants lost to follow-up at 12 months did not significantly differ in baseline characteristics or measures (age, sex, WAD grade II-III, level of pain, self-reported work ability, neck disability score and pain disability measure, p > 0.21).

Intervention

Between-group differences for the WAI at 3, 6 and 12 months are shown in Table III. The NSEB intervention was significantly better in improving work ability than the PPA intervention immediately after the intervention (3 months, p=0.03) and at all the follow-ups (6 months, p=0.01; 12 months, p=0.01). However, the NSEB intervention only showed significant improvement in work ability compared with the NSE intervention at 12 months (p=0.01). There was a significant difference between the NSE and PPA interventions only at 6 months.

Within-group differences for work ability demonstrated that participants in the NSEB intervention significantly improved work ability from baseline to

Table I. Baseline background characteristics of participants

	NSE		PPA	
Characteristics	(<i>n</i> = 60)	NSEB (n = 57)	(<i>n</i> = 48)	<i>p</i> -value ^a
Sex (female), n (%)	44 (73.3)	38 (66.7)	25 (52.1)	0.07
Age, years, mean (SD)	39 (11.7) ^b	40 (11.5) ^b	44 (9.4)	0.03
Months since injury, mean (SD)	19 (8.1)	20 (9.0)	19 (10.5)	0.64
WAD grade 2/3, n (%)	39 (65.0)/21 (35.0) ^c	27 (47.4)/30 (52.6)	29 (60.4)/19 (39.6)	0.14
Level of education, n (%)				0.66
High-school	31 (51.7)	34 (59.6)	24 (50)	
University	23 (38.3)	15 (26.3)	18 (37.5)	
Other	6 (10.0)	8 (14.0)	6 (12.5)	
Smoker (yes), n (%)	12 (20.0)	5 (8.8)	9 (18.8)	0.21
Pain medication (yes), n (%)	32 (53.3)	20 (35.1)	16 (33.3)	0.29
Employment status, full-time, n (%)	33 (66.0)	31(64.6)	28 (71.8)	0.76
Unsettled insurance claim, n (%)	15 (25.0)	12 (23.5)	10 (20.8)	0.88
Treatment received after WAD (yes), n (%)	51 (86.4)	35 (63.6)	32 (66.7)	0.01
Change of work task (yes), n (%)	11 (19.0)	9 (16.1)	7 (14.6)	0.84
Neck load at work, n (%)				0.71
Light	12 (20)	10 (18.1)	5 (10.9)	
Moderate	31 (51.7)	32 (58.1)	27 (58.7)	
Неаvy	17 (28.3)	13 (23.6)	14 (30.4)	
PDI, mean (SD)	23 (16)	19 (12)	19 (13)	0.27
HAD-D, mean (SD)	4.5 (3.9)	4.8 (4.4)	4.7 (4.2)	0.95
Financial situation, n (%)				0.17
Good	37 (55.2)	24 (44.4)	29 (60.4)	
Normal	14 (20.9)	21 (38.9)	16 (33.3)	
Bad	16 (28.1)	24 (44.4)	13 (27.1)	

^aTests conducted were: t-test for continuous characteristics and χ^2 test for categorical characteristics. ^bNeck-specific exercise group (NSE) and neck-specific exercise group with behavioural approach (NSEB) significantly different to prescription of physical activities (PPA). ^cNSE significantly different to NSEB and PPA.

SD: standardized deviation; PDI: Pain Disability Index; HAD-D: Hospital Anxiety and Depression Scale - depression.

Significant values are given in bold.

Table II. Work Ability Index score: raw values and percentage change from baseline

at each time-point			
Time-point	NSE	NSEB	PPA
Baseline	35.3	36	34.9
3 months	36.87 (4.5% increase)	37.04 (2.9% increase)	34.28 (1.8% decrease)
6 months	37.6 (6.5% increase)	37.19 (3.3% increase)	33.64 (3.6% decrease)
12 months	37.07 (5.0% increase)	38.73 (7.6% increase)	33.49 (4.1% decrease)

NSE: neck-specific exercise group; NSEB: neck-specific exercise group with behavioural approach; PPA: prescription of physical activities.

Table III Differences in Work Ability Index between arouns over time^a

			greepe ere					
NSE to PPA			NSEB to PPA			NSEB to NSE		
Difference	95% CI	<i>p</i> -value	Difference	95% CI	<i>p</i> -value	Difference	95% CI	<i>p</i> -value
1.64	-0.90-4.17	0.21	1.76	-0.55-4.07	0.14	0.89	-1.56-3.34	0.48
-0.15	-2.40-2.10	0.90	2.93	0.35-5.50	0.03	1.29	-1.07-3.65	0.08
2.55	0.01-5.09	0.05 ^b	3.44	0.86-6.01	0.01	1.91	-0.22-4.03	0.29
5.74	-3.66-15.13	0.23	12.31	2.82-21.80	0.01	6.58	1.40-11.76	0.01
	NSE to PPA Difference 1.64 -0.15 2.55 5.74	NSE to PPA Difference 95% CI 1.64 -0.90-4.17 -0.15 -2.40-2.10 2.55 0.01-5.09 5.74 -3.66-15.13	NSE to PPA Difference 95% CI p-value 1.64 -0.90-4.17 0.21 -0.15 -2.40-2.10 0.90 2.55 0.01-5.09 0.05 ^b 5.74 -3.66-15.13 0.23	NSE to PPA NSEB to PPA Difference 95% CI p-value Difference 1.64 -0.90-4.17 0.21 1.76 -0.15 -2.40-2.10 0.90 2.93 2.55 0.01-5.09 0.05 ^b 3.44 5.74 -3.66-15.13 0.23 12.31	NSE to PPA P-value Difference 95% CI 1.64 -0.90-4.17 0.21 1.76 -0.55-4.07 -0.15 -2.40-2.10 0.90 2.93 0.35-5.50 2.55 0.01-5.09 0.05 ^b 3.44 0.86-6.01 5.74 -3.66-15.13 0.23 12.31 2.82-21.80	NSE to PPA NSEB to PPA Difference 95% CI p-value Difference 95% CI p-value 1.64 -0.90-4.17 0.21 1.76 -0.55-4.07 0.14 -0.15 -2.40-2.10 0.90 2.93 0.35-5.50 0.03 2.55 0.01-5.09 0.05 ^b 3.44 0.86-6.01 0.01 5.74 -3.66-15.13 0.23 12.31 2.82-21.80 0.01	NSE to PPA NSEB to PPA NSEB to PPA NSEB to NSE Difference 95% CI p-value Difference 95% CI p-value Difference 95% CI Difference Difference 0.14 0.89 -0.15 -2.40-2.10 0.90 2.93 0.35-5.50 0.03 1.29 2.55 0.01-5.09 0.05 ^b 3.44 0.86-6.01 0.01 1.91 5.74 -3.66-15.13 0.23 12.31 2.82-21.80 0.01 6.58	NSE to PPA NSEB to PPA NSEB to PPA NSEB to NSE Difference 95% CI p-value Difference 95% CI p-value Difference 95% CI 0.14 0.89 -1.56-3.34 -0.15 -2.40-2.10 0.90 2.93 0.35-5.50 0.03 1.29 -1.07-3.65 2.55 0.01-5.09 0.05 ^b 3.44 0.86-6.01 0.01 1.91 -0.22-4.03 5.74 -3.66-15.13 0.23 12.31 2.82-21.80 0.01 6.58 1.40-11.76

^aAnalysis between interventions after adjusting for age, sex, time, group imes time interaction, neck load at work, Pain Disability Index, Hospital Anxiety and Depression Scale – depression and financial situation. ^bNote: 0.05^b=0.045.

NSE: neck-specific exercise group; NSEB: neck-specific exercise group with behavioural approach; PPA: prescription of physical activities; 95% CI: 95% confidence interval.

Significant values are given in bold.

Table IV. Within-group changes in Work Ability Index over time for each group^a

	Baseline to 3 months		Baseline to 6 months			Baseline to 12 months			
	Change	95% CI	<i>p</i> -value	Change	95% CI	<i>p</i> -value	Change	95% CI	p-value
NSE	1.57	-0.11-3.24	0.07	2.30	-0.71-5.31	0.13	1.77	-0.06-3.60	0.06
NSEB	1.04	-0.66-2.75	0.23	1.19	-0.90-3.29	0.27	2.73	0.86-4.60	0.004
PPA	-0.62	-2.64-1.40	0.55	-1.26	-6.43-3.91	0.64	-1.41	-3.76-0.93	0.24

^aAnalysis between interventions after adjusting for age, sex, time, group × time interaction, neck load at work, Pain Disability Index, Hospital Anxiety and Depression Scale – depression and financial situation.

NSE: neck-specific exercise group; NSEB: neck-specific exercise group with behavioural approach; PPA: prescription of physical activities; 95% CI: 95% confidence interval.

Significant value is given in bold.

one year (p < 0.01), although no significant short-term improvement was observed. Only a trend of improvement over one year for participants in the NSE intervention was observed (Table IV). In contrast, those in the PPA intervention had a trend of decreasing work ability over time.

Associations

In preliminary testing, there were no significant interactions between group and age (group \times age, p=0.93) or sex (group \times sex, p=0.33). However, interaction between group and time was significant (p=0.03). An unstructured correlation structure was used in the modelling as it produced the lowest OIC value compared with other alternatives, such as independent or exchangeable. Table V displays the results of the final model for the interventions on WAI after controlling for significant variables and adjusting for age, sex, time and group × time interaction. Overall, group allocation was not significantly associated with the score on the WAI over one year (p=0.16). In the final predictive model, in order of magnitude of coefficients the variables associated with a decrease in WAI over one year were: moderate and heavy load on the neck at work, poorer financial situation, depression and greater painrelated disability (p < 0.01).

Intervention adherence

Adherence to the PPA intervention (50%) was significantly less than to the NSE (72%) and NSEB (80%) interventions (p=0.02) at 3 months. However, the percentage of participants who had completed more than half of their prescribed exercise during the maintenance phase was similar at 6 months (p=0.55) and 12 months (p=0.26). At 6-month follow-up, participation in exercise was 42%, 51% and 48% in the NSE, NSEB and PPA groups, respectively. At 12 months,

Table V. Factors associated with changes in Work Ability Index (WAI) over one year using generalized estimating equations (GEE)^a

Predictor variable	Coefficient ^b	95% CI	<i>p</i> -value
Neck load at work			< 0.001
Light (control)	-	-	-
Moderate	-2.70	-4.69 to -0.69	0.01
Heavy	-4.91	-7.13 to -2.69	< 0.001
Financial situation			< 0.001
Good (control)	-	-	-
Normal	-3.07	-4.63 to -1.51	< 0.001
Bad	-3.89	-6.37 to -1.42	0.002
PDI	-0.18	-0.24 to -0.11	< 0.001
HAD-D	-0.38	-0.59 to -0.17	< 0.001

^aAdjusted with age, sex, time, group×time interaction.

^bCoefficient is the unstandardized regression coefficient

CI: confidence Interval; PPA: prescription of physical activities; NSE: neckspecific exercise group; NSEB: neck-specific exercise group with behavioural approach; PDI: Pain Disability Index, HAD-D: Hospital Anxiety and Depression Scale – depression; 95% CI: 95% confidence interval. participation in the NSE group was still lower (26%) than in the NSEB group (33%) and both were higher than the PPA group (18%).

DISCUSSION

This study investigated the efficacy of 3 different exercise interventions on self-reported work ability for individuals with chronic WAD grade II–III and which baseline characteristics were associated with the WAI over time. The results show that NSEB intervention was significantly better than PPA intervention for improving work ability. An improving trend in work ability was found only in the NSE and NSEB interventions, with participants in the NSEB intervention having significantly improved work ability after one year. Of concern was the trend of decreasing work ability in those participating in the PPA intervention. Overall, the NSEB intervention was more efficacious in improving self-reported work ability than the NSE and PPA interventions.

The conceptual background of work ability is based on the stress-strain concept and balance model, in which a balance between a person's resources and work demands is required to safeguard health (29). The finding that the overall mean baseline self-reported work ability of the participants in this study was 35.4, indicating moderate work ability, suggests that individuals with WAD are under some strain requiring additional support. This score is higher than workers with chronic neck pain and those on long-term sick leave (mean score 19.1) (30). This is not unexpected, as only those who were employed completed the WAI, and they would have higher work ability than those on sick leave.

These results support our hypothesis that NSEB is superior in improving self-reported work ability compared with the other 2 exercise interventions tested (NSE and PPA) despite similar adherence during the maintenance phase at 6- and 12-month follow-ups. The addition of a behavioural component may have allowed patients to problem-solve their symptoms during a relapse, learn skills to manage pain with relaxation exercises, and succeed in exercise progression. It is recognized that work ability is related to both condition-specific and psychosocial factors for individuals with chronic WAD (31). The addition of the behavioural approach may address the psychosocial risk factors, similar to that of a psychosocial risk reduction intervention, which is suggested to be effective in improving function and return to work for people with WAD (32). Our results are similar to a study that showed that exercises with a cognitive behavioural approach improved self-evaluated work ability in workers with low back pain (33).

The results suggest that the NSE intervention is more efficacious in improving work ability than is PPA, but only at 6 months. A possible explanation for this is that NSE alone was insufficient to improve work ability, as work ability is dependent on the interactions between organizational factors involving the work environment, work content and demands, the work community, and leadership (34). The NSEB intervention was only statistically significant in improving work ability at one year compared with baseline, with an increase of at least 3 points previously accepted as "improved work ability" (35). This suggests that behavioural skills might take time to develop and master.

The PPA group showed a trend of decreasing work ability over time, even though physical exercise has been shown to maintain work ability for workers in the metal and retail industry (36). This may be due to lower participation in the PPA group than either of the neck-specific exercise groups during the intervention period and at 12 months, or that general exercise alone is insufficient to create change in chronic WAD. This finding suggests that the chronic WAD population is at risk of decreasing work ability if their concerns are not addressed, and increasing physical activity alone may not be effective in improving work ability. In the final model time was not significantly associated with work ability, indicating that work ability does not improve over time without specific interventions in individuals with chronic WAD. As self-reported work disability is less likely to change over time for individuals with WAD, early intervention should be implemented to maintain and improve work ability even if workers have returned to work. This study also shows that existing work ability for individuals with chronic WAD grades II and III can be improved with appropriate treatment.

The following baseline characteristics: increased perceived neck load at work, greater disability due to pain, greater depression and poorer self-rated financial situation were significantly associated with a reduction in WAI over one year. Similar findings have been reported in the literature for populations with WAD, in that a combination of work-related, pain-related, psychological and individual variables is prognostic for work-related outcomes (31).

Several individual and work-related factors were associated with work ability for individuals with WAD, of which some were consistent with a recent review (12). Increased perceived physical neck load at work was significantly associated with poorer work ability. This is consistent with previous reports that high physical demands at work may impair work ability (35). As the majority of participants in this study (83%) reported moderate to heavy physical neck load during work, it is possible that delivering an NSEB intervention with ergonomic advice on safe work procedures may achieve additional gains in work ability. At an organizational level, employers can contribute to improving work ability through ergonomic interventions, as workplace ergonomics can significantly influence an individual's work ability (35). Greater disability due to pain, as indicated by the Pain Disability Index, was one of the significant contributors to poorer work ability (37). It is possible that greater self-rated disability in normal life would translate to self-rated disability in work. This is consistent with the literature that greater disability is linked to increased recovery time (1). Our findings indicate that depression was significantly associated with poorer work ability. Consistent with the literature, workers with depression in the general population are at greater risk of having lower work ability in the long term (38). Depression is common in individuals with chronic WAD, thus mental health support may be needed for this population. Poorer self-rated financial situation was also found to be significantly associated with decreased work ability. Perceived financial situation is linked with mental health and perceived employment commitment (39).

This study has several strengths. It is the first to report the efficacy of an exercise intervention on work ability for individuals with chronic WAD grades II and III. In addition, the generalizability of our results is enhanced by the fact that participants were randomly allocated to the different interventions, conducted in multiple primary care centres.

Several limitations may have influenced the findings of this study. Firstly, it is a secondary analysis of a randomized controlled trial, which was not powered to detect a change in the WAI. Only those who were currently employed, or were employed prior to their WAD injury, completed the WAI questionnaire, which may explain the relatively high baseline WAI score and limited sample size. Secondly, this was an exploratory study, as a large range of baseline variables with the potential to influence work ability were included. However, controlling for the significant variables increases our confidence that any changes observed are probably due to the efficacy of the intervention. Participation in the intervention was reasonable during the active period, but reduced substantially over time. It is possible that stronger results would have been noted with better participation, although no minimum threshold has been established. The loss to follow-up over the 12-month trial varied from 24% to 30%, which may bring the validity of the results into question. However, at 12 months, the majority (>63%) of participants remained in the study. It is possible that other factors not assessed in this study may have contributed to the changes observed. For example, Tuomi et al. (35) found that reduced management strain and improved work postures and greater satisfaction with the supervisor's attitude was associated with improved work ability over an 11-year period, while Emberland et al. (23) found that role conflict and human resource primacy and positive challenge were most predictive of work ability in a 2-year prospective study. Nevertheless, the study is novel; the results provide insight into how best to improve work ability for individuals with chronic WAD. Further research is needed into how best to support individuals with chronic WAD to remain at work in the short term and how best to support those on sick-leave. Interventions that include the workplace, such as enhancing the individual's self-management skills to make adjustments at work, and targeting the supervisors, may promote staying at work (37). This approach is consistent with the multidimensional model of work ability that emphasizes the importance of having control over one's work and the way in which it is organized (40).

In summary, a NSEB intervention was better at improving self-reported work ability than NSE alone or PPA. The results of this study suggest that improvement in work ability is associated with work-related, pain-related, psychological and individual factors. A NSEB intervention shows promise for improving work ability in those remaining at work with a WAD grade II–III injury. Further research is needed to identify interventions that will be effective in promoting work ability in the short- and long-term in this population.

ACKNOWLEDGEMENTS

The authors thank all the participants, physical therapists and staff involved in any stage of the study.

The Regional Ethics Committee at the Medical Faculty at Linköping University in Sweden provided ethics and has granted approval for the protocol (2010/18888-31 and 2011/262-32).

The study was funded by the Swedish government through the Swedish Research Council, in cooperation with the Swedish Social Insurance Agency through the REHSAM Foundation Stockholm, Sweden, and the regional Centers for Clinical Research and the County Council of Östergötland, Linköping, Sweden and Sörmland, Uppsala, Sweden Research Council of Southeast Sweden.

REFERENCES

- Carroll LJ, Holm LW, Hogg-Johnson S, Côté P, Cassidy D, Haldeman S, et al. Course and prognostic factors for neck pain in whiplash-associated disorders (WAD): results of the Bone and Joint Decade 2000–2010 Task Force on Neck Pain and Its Associated Disorders. Spine 2008; 33: S83–S92.
- Edwards M, Smith S, Zuby DS, Lund AK, editors. Improved seat and head restraint evaluations. Proceedings of the 19th International Technical Conference on Enhanced safety of Vehicles; 2005.
- 3. Carlsson I, Hedin M, Brorsson J, Gårdestig S, Lundgren M,

Rehnqvist N, et al. The Whiplash Commission Final Report. Stockholm: Whiplashkommissionen 2005.

- Styrke J, Stålnacke B-M, Bylund P-O, Sojka P, Björnstig U. Neck injury after whiplash trauma in a defined population in northern Sweden – long-term sick leave and costs of low productivity. Epidemiol 2014; 4: 1–8.
- Biering-Sorensen S, Moller A, Stoltenberg CD, Holm JW, Skov PG. The return-to-work process of individuals sicklisted because of whiplash-associated disorder: a threeyear follow-up study in a Danish cohort of long-term sickness absentees. BMC Public Health 2014; 14: 113.
- Leth-Petersen S, Rotger GP. Long-term labour-market performance of whiplash claimants. J Health Econ 2009; 28: 996–1011.
- Holm LW, Cassidy D, Sjogren Y, Nygren D. Impairment and work disability due to whiplash injury following traffic collisions, an analysis of insurance material from the Swedish Road Traffic Injury Commission. Scand J Public Health 1999; 27: 116–123.
- Carriere JS, Thibault P, Adams H, Milioto M, Ditto B, Sullivan MJL. Expectancies mediate the relationship between perceived injustice and return to work following whiplash injury: a 1-year prospective study. Eur J Pain 2017; 21: 1234–1242.
- Kamper SJ, Rebbeck TJ, Maher CG, McAuley JH, Sterling M. Course and prognostic factors of whiplash: a systematic review and meta-analysis. Pain 2008; 138: 617–629.
- Buitenhuis J, de Jong PJ, Jaspers JC, Groothoff JW. Work disability after whiplash, a prospective cohort study. Spine 2009; 34: 262–267.
- Elfving B, Asell M, Luning Bergsten C, Alexanderson K. Exploring activity limitations and sick leave among patients with spinal pain participating in multidisciplinary rehabilitation. Disabil Rehabil 2010; 32: 292–299.
- van den Berg TI, Elders LA, de Zwart BC, Burdorf A. The effects of work-related and individual factors on the Work Ability Index: a systematic review. Occup Environ Med 2009; 66: 211–220.
- Amirfeyz R, Cook J, Gargan M, Bannister G. The role of physiotherapy in the treatment of whiplash associated disorders: a prospective study. Arch Orthop Trauma Surg 2009; 129: 973–977.
- Baltov P, Cote J, Truchon M, Feldman DE. Psychosocial and socio-demographic factors associated with outcomes for patients undergoing rehabilitation for chronic whiplash associated disorders: a pilot study. Disabil Rehabil 2008; 30: 1947–1955.
- Sullivan M, Adams H, Thibault P, Moore E, Carriere JS, Lariviere C. Return to work helps maintain treatment gains in the rehabilitation of whiplash injury. Pain 2017; 158: 980–987.
- Kongsted A, Qerama E, Kasch H, Bendix T, Bach FW, Korsholm L, et al. Neck collar, "act-as-usual" or active mobilization for whiplash injury? A randomized parallelgroup trial. Spine (Phila Pa 1976) 2007; 32: 618–626.
- Crawford JR, Khan RJ, Varley GW. Early management and outcome following soft tissue injuries of the neck-a randomised controlled trial. Injury 2004; 35: 891–895.
- Rosenfeld M, Seferiadis A, Gunnarsson R. Active involvement and intervention in patients exposed to whiplash trauma in automobile crashes reduces costs: a randomized, controlled clinical trial and health economic evaluation. Spine (Phila Pa 1976) 2006; 31: 1799–1804.
- Bunketorp L, Lindh M, Carlsson J, Stener-Victorin E. The perception of pain and pain-related cognitions in subacute whiplash-associated disorders: its influence on prolonged disability. Disabil Rehabil 2006; 28: 271–279.
- Ludvigsson ML, Peterson G, O'Leary S, Dedering A, Peolsson A. The effect of neck-specific exercise with, or without a behavioral approach, on pain, disability, and self-efficacy in chronic whiplash-associated disorders: a randomized clinical trial. Clin J Pain 2015; 31: 294–303.
- 21. Peolsson A, Landén Ludvigsson M, Overmeer T, Dedering

Å, Bernfort L, Johansson G, et al. Effects of neck-specific exercise with or without a behavioural approach in addition to prescribed physical activity for individuals with chronic whiplash-associated disorders: a prospective randomised study. BMC Musculoskelet Di 2013; 14: 311.

- Landén Ludvigsson M, Peolsson A, Peterson G. Neckspecific exercise program: Linköping: Linköping University Electronic Press; 2015. Available from: http://urn.kb.se/ resolve?urn=urn:nbn:se:liu:diva-113865.
- Emberland JS, Knardahl S. Contribution of psychological, social, and mechanical work exposures to low work ability: a prospective study. J Occup Environ Med 2015; 57: 300–314.
- Tuomi K, Toikkanen J, Eskelinen L, Backman AL, Ilmarinen J, Jarvinen E, et al. Mortality, disability and changes in occupation among aging municipal employees. Scand J Work Environ Health 1991; 17: 58–66.
- Ilmarinen J. The Work Ability Index (WAI). Occup Med 2007; 57: 160.
- Aittomaki A, Lahelma E, Roos E. Work conditions and socioeconomic inequalities in work ability. Scand J Work Environ Health 2003; 29: 159–165.
- Dzewaltowski DA. Toward a model of exercise motivation. J Sport Exercise Psy 1989; 11: 251–269.
- Bassett DR. International physical activity questionnaire: 12-country reliability and validity. Med Sci Sport Exer 2003; 35: 1396.
- Ilmarinen J, Tuomi K, Eskelinen L, Nygard CH, Huuhtanen P, Klockars M. Background and objectives of the Finnish research project on aging workers in municipal occupations. Scand J Work Environ Health 1991; 17 Suppl 1: 7–11.
- 30. Dellve L, Ahlstrom L, Jonsson A, Sandsjö L, Forsman M, Lindegård A, et al. Myofeedback training and intensive muscular strength training to decrease pain and improve work ability among female workers on long-term sick leave with neck pain: a randomized controlled trial. Int Arch Occup Environ Health 2011; 84: 335–346.
- 31. Agnew L, Johnston V, Landen Ludvigsson M, Peterson G,

Overmeer T, Johansson G, et al. Factors associated with work ability in patients with chronic whiplash-associated disorder grade II–III: a cross-sectional analysis. J Rehabil Med 2015; 47: 546–551.

- 32. Ehrenborg C, Gustafsson S, Archenholtz B. Long-term effect in ADL after an interdisciplinary rehabilitation programme for WAD patients: a mixed-method study for deeper understanding of participants' programme experiences. Disabil Rehabil 2014; 36: 1006–1013.
- Suni J, Rinne M, Natri A, Statistisian MP, Parkkari J, Alaranta H. Control of the lumbar neutral zone decreases low back pain and improves self-evaluated work ability: a 12-month randomized controlled study. Spine 2006; 31: E611–E620.
- Ilmarinen J, Tuomi K. Past, present and future of work ability. Helsinki, Sweden: Finnish Institute of Occupational Health, 2004 Contract No.: People and Work Research Reports no 65.
- Tuomi K, Ilmarinen J, Martikainen R, Aalto L, Klockars M. Aging, work, life-style and work ability among Finnish municipal workers in 1981–1992. Scand J Work Environ Health 1997; 23 suppl 1: 58–65.
- Tuomi K, Vanhala S, Nykyri E, Janhonen M. Organizational practices, work demands and the well-being of employees: a follow-up study in the metal industry and retail trade. Occup Med 2004; 54: 115–121.
- De Vries HJ, Reneman MF, Groothoff JW, Geertzen JHB, Brouwer ML. Factors promoting staying at work in people with chronic nonspecific musculoskeletal pain: a systematic review. Disabil Rehabil 2012; 34: 443–458.
- Gilmour H, Patten SB. Depression and work impairment. Health Reports 2007; 18: 9.
- Rantakeisu U, Jönsson LR. Unemployment and mental health among white-collar workers – a question of work involvement and financial situation? Int J Soc Welf 2003; 12: 31–41.
- 40. Ilmarinen J, Tuomi K, Seitsamo J. New dimensions of work ability. Int Congr Ser 2005; 1280: 3–7.